Author’s response to reviews

Title: Seroprevalence of Transfusion Transmitted Infections among Blood Donors in Gash Barka Zonal Blood Transfusion Center, Barentu, Eritrea, 2014 through 2017

Authors:

Yacob Tesfamichael Keleta (jacobcls2017@gmail.com;yacobtesfamichael@yahoo.com)
Oliver Okoth Achila (oliverachila@gmail.com)
Absera Woldu Haile (absir26@gmail.com)
Bereket Habteslasie Gebrecherkos (berekethabteslasie112@gmail.com)
Danait Tareke Tesfaldet (danaittareke@gmail.com)
Kibrom Solomon Teklu (kibromsolomon58@gmail.com)
Mesuda Abrhum Mohammed (mesuedaabruhum@gmail.com)
Selihom Tesfaslase Ghedel (selihomtesfaslase@gmail.com)

Version: 2 Date: 24 Jan 2019

Author’s response to reviews:

Dear Editor

This is to confirm that we have duly responded to the suggestions raised by your reviewers and editorial team. The highlighted concerns included the following:

Requested revisions

Editor comments: Please check throughout your manuscript and correct grammatical errors.

✧ We have thoroughly checked the manuscript and made necessary grammatical amendments.
Reviewer 2 comments: The total seropositivity of TTIs in table 2 is different from those of other tables (265 vs 266). There are still grammatical errors.

✧ The correct total seropositivity of TTIs is 266. The total TTIs in table 2 was wrong, the reason is that the positive tests for syphilis for the year 2017 (3) was wrong, sidewise the total positivity for syphilis (139) was also wrong. Therefore the correct figures are 4 for the year 2017 and 140
for total syphilis positivity.

We have made necessary grammatical corrections.